Author's response to reviews

Title: "Taking Your Place at the Table": The Value of Chaplains as Integral Members of Interdisciplinary Research Teams

Authors:

Allison Kestenbaum (alkestenbaum@jtsa.edu)
Jennifer James (jennifer.james@ucsf.edu)
Stefana Borovska (stefana.borovska@ucsf.edu)
Michele Shields (michelershields@me.com)
Will Hocker (will.hocker@ucsfmedctr.org)
Michael Rabow (mrabow@medicine.ucsf.edu)
Laura B. Dunn (laura.dunn@ucsf.edu)

Version: 2 Date: 30 July 2014

Author's response to reviews:

We have made all of the requested changes that were editorial in nature.
1. We have updated the title page to include all of the requested information.
2. We have added a "Competing Interests" section after the Conclusions.
3. We have added an "Acknowledgements" section after the Authors' Contributions.
4. We have added the requested ethics information.
5. We have added the requested information about informed consent.
6. We have added a separate "Conclusions" section after the Discussion.
7. We have removed the legends from the figure images, and have added a Figure Legends section after the References section.
8. We have revised the manuscript to include line numbers.